BMJ Open Pathway Of Low Anterior Resection syndrome (LARS) relief after Surgery (POLARiS): protocol for an international, open-label, multi-arm, phase 3 randomised superiority trial within a cohort, with economic evaluation, process evaluation and qualitative sub-study, to explore the natural history of LARS and compare transanal irrigation and sacral neuromodulation to optimised conservative management for people with major LARS following a high or low anterior resection for colorectal cancer

To cite: Croft J. Farrow E. Coxon-Meggy AH, et al. Pathway Of Low Anterior Resection syndrome (LARS) relief after Surgery (POLARiS): protocol for an international, open-label, multi-arm, phase 3 randomised superiority trial within a cohort, with economic evaluation, process evaluation and qualitative sub-study, to explore the natural history of LARS and compare transanal irrigation and sacral neuromodulation to optimised conservative management for people with major LARS following a high or low anterior resection for colorectal cancer. BMJ Open 2025;15:e092612. doi:10.1136/ bmjopen-2024-092612

Prepublication history and additional supplemental material for this paper are available online. To view these files, please visit the journal online (https://doi.org/10.1136/ bmjopen-2024-092612).

Received 19 August 2024 Accepted 29 November 2024



@ Author(s) (or their employer(s)) 2025. Re-use permitted under CC BY. Published by BMJ Group.

For numbered affiliations see end of article.

Correspondence to

Julie Cornish: julie.cornish@wales.nhs.uk Julie Croft , ¹ Emily Farrow , ^{2,3} Alexandra Harriet Coxon-Meggy , ³ Katie Gordon , ¹ Neil Corrigan , ¹ Hannah Mather, ¹ Deborah D Stocken, ¹ Megan Dale , Heir Corngan , Hannan Mather, Deboran D Stocken, Megan Dale , Huey Yi Chong, Judith White, Laura Knight, Alun Meggy, Christina Lloydwin, Betty Tan, Ashley Douglas, Ralph Powell, Julie Hepburn, David Jayne, Jared Torkington , Andrea Warwick, Kheng-Seong Ng, Andrea Warwick, Kheng-Seong Ng, Aaron Quyn, Aaron Quyn, Julie Cornish

ABSTRACT

Introduction As a result of improving survival rates, the adverse consequences of rectal cancer surgery are becoming increasingly recognised. Low anterior resection syndrome (LARS) is one such consequence and describes a constellation of bowel symptoms after rectal cancer surgery which includes urgency, faecal incontinence, stool clustering and incomplete evacuation. LARS has a significant adverse impact on quality of life (QoL) and symptoms are present in up to 75% of patients in the first year after surgery. Despite this, little is known about the natural history and there is poor evidence to support current treatment options. Methods and analysis The objectives of POLARiS are to explore the natural history of LARS and to evaluate the clinical and cost-effectiveness of transanal irrigation (TAI) or sacral neuromodulation (SNM) compared with optimised conservative management (OCM) for people with major LARS.

STRENGTHS AND LIMITATIONS OF THIS STUDY

- ⇒ The trial is pragmatically designed to optimise and assess recruitment and retainment.
- ⇒ This trial includes an economic evaluation of treatment options specific to both the UK and Australia.
- ⇒ Lay representatives with personal experience of bowel cancer and low anterior resection syndrome (LARS) have contributed throughout the trial design and ongoing Trial Management Group meetings.
- ⇒ There are recognised potential limitations to the LARS score, including limited sensitivity to detect real-time change in response to treatment. Additional outcome measures of quality of life and a new LARS patient-reported outcome measure are being collected to give a more nuanced picture of treatment response.

POLARiS is a prospective, international, open-label, multi-arm, phase 3 randomised superiority trial within a cohort design, with internal pilot phase, qualitative sub-study, process evaluation and economic evaluation. Approximately 1500 adult participants from UK hospitals and 500 from Australian hospitals who have undergone a high or low anterior resection for colorectal cancer in the last 10 years will be recruited into the cohort. Six-hundred participants from the UK and 200 participants from Australia, with major LARS symptoms, defined as a LARS score of ≥30, will be recruited to the randomised controlled trial (RCT) element. Participants entering the RCT will be randomised between OCM, TAI or SNM, all with equal allocation ratios.

Cohort and RCT participants will be followed up for a 24-month period, completing a series of questionnaires measuring LARS symptoms and QoL, as well as clinical review for those in the RCT. A process evaluation, qualitative sub-study and economic evaluation will also be conducted. The primary outcome measure of the POLARIS cohort and RCT is the LARS score up to 24 months post-registration/randomisation. Analyses of the RCT will be conducted on an intention-to-treat basis. Comparative effectiveness analyses for each endpoint will consist of two pairwise treatment comparisons: TAI versus OCM and SNM versus OCM. Secondary outcomes include health-related QoL, adverse events, treatment compliance and cost-effectiveness (up to 24 months post-registration/randomisation).

Ethics and dissemination Ethical approval has been granted by Wales REC 4 (reference: 23/WA/0171) in the UK and Sydney Local Health District HREC (reference: 2023/ETH00749) in Australia. The results of this trial will be disseminated to participants on request and published on completion of the trial in a peer-reviewed journal and at international conferences.

Trial registration number ISRCTN12834598; ACTRN12623001166662.

INTRODUCTION

Colorectal cancer is the fourth most common cancer in the UK, with an average of 44100 new cases diagnosed per year. Of these, around 30% of patients will have rectal cancer, making it the most common site of colorectal cancer. Many of these patients will undergo major resection. With evolving surgical techniques and the development of adjuvant and neoadjuvant treatments the 5-year survival rate for colorectal cancer has improved by around 30% over four decades. 3

As a result of improving survival rates, the adverse consequences of rectal cancer surgery are becoming increasingly recognised. Low anterior resection syndrome (LARS), initially defined in 2012, is one such consequence and describes a constellation of bowel symptoms after rectal cancer surgery which includes urgency, faecal incontinence, stool clustering and incomplete evacuation. LARS has a significant adverse impact on quality of life (QoL). LARS symptoms are present in up to 75% of patients in the first year after surgery, remaining in up to half of these patients over 10 years. Due to the significant burden and persistence of LARS symptoms, there is an urgent need to improve the understanding of the natural history and treatment for LARS.

Conservative management of LARS includes dietary modifications, medication and physiotherapy. Dietary modifications are hugely variable with many patients implementing their own trial-and-error strategies.⁸ Input from dieticians may help to meet this challenge, but often these services are not readily available.⁹ Pelvic floor physiotherapy

and biofeedback techniques have been reported to improve symptoms, although only a few low-powered studies have been published. These services are not readily available at all hospitals and can have prolonged waiting times. Overall, there is little structured guidance on conservative management with a high degree of variability in what centres offer. Conservative treatment for LARS largely targets faecal incontinence symptoms; however, many patients present with predominantly obstructive symptoms such as emptying difficulties and tenesmus. Little is known about whether there is a difference in treatment efficacy in patients presenting with a different cluster of symptoms.

For those patients who fail conservative management or suffer from major LARS (LARS score≥30), treatment options include transanal irrigation (TAI) and sacral neuromodulation (SNM). Several small randomised controlled trials (RCTs) and studies have assessed the efficacy of TAI on bowel function after low anterior resection, but the sample sizes are small with variable uptake and significant heterogeneity. There is variable use of TAI after surgery and the timing, with less than one-third of surgeons considering TAI as an option for the treatment of LARS. 16 Currently, SNM is only licenced for use in faecal incontinence, but a recent meta-analysis of 10 studies suggests that significant improvements in function might be achieved in LARS; however, these conclusions are limited by small sample size and significant heterogeneity between studies. 17 Most recently, the SANLARS RCT showed both symptom and QoL improvement at 6- and 12-month follow-ups in patients with major LARS. 18

As patients are increasingly surviving rectal cancer surgery, it is essential we examine the impact and consequences on patient's lives and define treatment pathways that are currently lacking robust evidence from RCTs. The National Institute for Clinical Excellence has developed a research recommendation to determine the effectiveness and safety of SNM and TAI compared with symptomatic treatment for people with major LARS. 19 As very few studies report beyond 12 months, we also lack knowledge of the natural history of LARS. Is there a time point where LARS symptoms become stable? If we delay treatment postoperatively are symptoms managed as effectively? Is there a difference in long-term outcomes in patients who have had radiotherapy? Understanding these may allow us to make more informed decisions regarding the initiation of treatment and be able to better advise patients on how their symptoms and treatment may change over time. POLARiS is a large-scale, complex randomised superiority trial that will address these research priorities.

OBJECTIVES

- ► The primary objective of the POLARiS cohort is to explore the natural history of LARS over time.
- ➤ The primary objective of the POLARIS RCT is to evaluate the clinical and cost-effectiveness of TAI or SNM compared with optimised conservative management (OCM) for people with major LARS.

Figure 1 POLARIS trial schema, LARS iCAT, low anterior resection syndrome Impact and Consequences Tool: OCM, optimised conservative management; SNM, sacral neuromodulation; TAI, transanal irrigation.

METHODS & ANALYSIS Study design

POLARiS is a prospective, international, open-label, multi-arm, phase 3 randomised superiority trial within a cohort²⁰, with internal pilot phase, qualitative sub-study process and economic evaluation (figure 1). The Clinical Trials Research Unit (CTRU) at the University of Leeds will perform co-ordination of the trial with Cardiff & Vale University Health Board (C&V UHB) as the UK sponsor and the University of Sydney as the Australian sponsor.

Patients, aged ≥18 years who have had a high or low anterior resection within 10 years for rectal cancer with a functioning anastomosis and meet the eligibility criteria will be invited to the cohort (table 1). Patients with major LARS and who meet the eligibility criteria will be invited to take part in the RCT. Participants in the cohort and RCT will be asked to complete a set of baseline questionnaires. At 3 monthly intervals after registration/randomisation participants will complete a LARS score and a series of QoL Questionnaires. The RCT treatments are OCM, TAI and SNM. Neither clinicians nor participants will be blinded to the RCT treatment allocation.

Study population

A total of 1500 UK participants and 500 Australian participants will be entered into the cohort with the aim to randomise 600 from the UK and 200 from Australia into the RCT. Each site must fulfil a set of pre-specified criteria, and the local principal investigator (PI) must complete a form to verify the site is willing and able to comply with trial requirements. The trial will open in at least 20 UK and 15 Australian hospitals.

Participating sites must be able to offer at least one of the interventions (TAI or SNM), perform at least 30 anterior resections for colorectal cancer each year and at least 10 SNM procedures each year if delivering SNM to trial participants. Sites which offer TAI and have an existing Service Level Agreement with a centre offering SNM, have the option for three-way randomisation.

Recruitment

Patients will be screened through cancer databases, clinical notes and clinics at participating hospital sites. Those who meet the cohort eligibility criteria (table 1) will be provided with a cohort participant information sheet (PIS). Once written informed consent is obtained, participants are registered using a central web-based 24-hour registration system (online supplemental file 1) Cohort Consent Form.

Participants identified as having a major LARS score (LARS score≥30), documented within the last 3 months, at the point of screening and who meet the RCT eligibility criteria (table 1) can be considered for direct entry into the RCT without registration into the cohort. Participants from the cohort who subsequently become eligible for the RCT (eg, identification of a major LARS score through completion of the cohort LARS questionnaire) will be invited to participate in the RCT. Eligible patients will be provided with the RCT PIS and will discuss the trial with a suitably qualified member of the healthcare team. Participants will be given as much time as required (ideally a minimum of 24 hours) to consider the information before written informed consent is taken. Randomisation will

Table 1 Eligibility crite	eria for both cohort and randomised controlled trial	participants
	Cohort eligibility criteria	RCT eligibility criteria
Inclusion criteria	 Aged ≥18 years Able to provide written informed consent Diagnosis of rectal or sigmoid cancer Low or high anterior resection (colorectal resection with anastomosis to the rectum) Functioning anastomosis Primary surgery less than 10 years before recruitment At least 6 months since reversal of stoma or primary surgery if no stoma created Able and willing to comply with the terms of the protocol including participant completed questionnaires 	 Aged ≥18 years Able to provide written informed consent Diagnosis of rectal or sigmoid cancer Low or high anterior resection (colorectal resection with anastomosis to the rectum) Functioning anastomosis Primary surgery less than 10 years before recruitment At least 6 months since reversal of stoma or primary surgery if no stoma created Able and willing to comply with the terms of the protocol including participant completed questionnaires Major LARS symptoms within the last 3 months (defined as a LARS score of ≥30) Clinically appropriate for randomisation as determined by treating clinician
Exclusion criteria	 Receiving ongoing chemotherapy, radiotherapy or immunotherapy treatment for cancer Anterior exenteration 	 Receiving ongoing chemotherapy, radiotherapy or immunotherapy treatment for cancer Metastatic disease Inflammatory bowel disease Pregnancy Use of TAI for LARS within 1 month before randomisation Not eligible for SNM and not eligible for TAI Anterior exenteration Anastomotic stricture History of anastomotic leak with evidence of ongoing leak/sinus
		Exclusion criteria for SNM: ➤ Site unable to offer SNM as a treatment ➤ Previous SNM ➤ Margin Positive (R1) resection within 24 months before randomisation ➤ Specific contraindications to implantation ➤ Any other contraindications advised by the care team, product manufacturer or distributor Exclusion criteria for TAI: ➤ Unable to perform TAI ➤ Previous use of TAI for LARS ➤ Site unable to offer TAI as a treatment ➤ Any other contraindications advised by the care team, product manufacturer or distributor

LARS, low anterior resection syndrome; OCM, optimised conservative management; RCT, randomised controlled trial; SNM, sacral intermodulation; TAI, transanal irrigation.

take place as soon as possible after written informed consent is obtained (online supplemental file 1) RCT Consent Form.

Randomisation

There are three randomisation options, dependent on the patient's eligibility (table 1) and the site's capacity to deliver TAI or SNM, each with equal allocation ratios, listed below:

- 1. OCM vs TAI vs SNM
- 2. OCM vs SNM
- 3. OCM vs TAI

Randomisation will be performed using the central automated web-based 24-hour randomisation system. Within each randomisation option a computer-generated minimisation programme incorporating a

random element^{21 22} will be used to allocate patients with a 1:1(:1) ratio, with the following minimisation factors: recruiting site, time from surgery, biological sex at birth, age, radiotherapy and procedure (high or low anterior resection).

Participants who were initially eligible for and randomised via the 3-way randomisation option, may be eligible for second randomisation if they do not respond to their first allocated treatment option. For participants who undergo a second randomisation, data will continue to be collected in line with primary randomisation timelines.

Patient and public involvement

A total of four lay representatives from the UK and Australia with relevant personal experience of

		Trial Time points (post-randomisation)								
	Events	Baseline ¹	3 months	6 months	9 months	12 months	Continue 3 monthly ²	24 months		
ts ns	Consent	✓								
ents	Randomisation	✓								
<u> </u>	Clinical Assessment	✓	✓							
ssessmen / vestigatio	Telephone Assessment			✓	✓	✓				
Assessi / Investig	Clinical Data Collection (medical notes)							✓		
	Adverse Events		✓	✓	✓	✓		✓		
v	LARS Questionnaire	✓	✓	✓	✓	✓	✓	✓		
Participant completed lestionnaires	LARS ICAT	✓	✓	✓		✓		✓		
	EQ-5D-5L	✓	✓	✓		✓		✓		
	EORTC CR29	✓	✓	✓		✓		✓		
	QLQ-C30	✓	✓	✓		✓		1		
ਰ	Health Resource Use	✓	✓	✓		✓		✓		

Interview Survey

Figure 2 The timings of data collection points and clinical assessments for the randomised controlled trial patients randomised to optimised conservative management are summarised in the table below. LARS iCAT, low anterior resection syndrome Impact and Consequences Tool.

colorectal cancer, LARS and SNM were recruited. Lay representatives contributed to trial development, production of the lay summary and have made active contributions in co-applicant group meetings to guide decisions and provide recommendations.

An additional focus group was conducted in June 2021 with 12 bowel cancer patients from the UK and Australia to investigate key aspects of the trial design covering key areas such as acceptability of randomisation and treatments. This resulted in the development of the three randomisation pathways, the option of a second randomisation and adjustments to the OCM pathway.

INTERVENTIONS

All RCT patients will be provided with a LARS Information Booklet, a detailed support document developed by experts specifically developed for use in the POLARiS trial. The booklet aims to inform and advise participants regarding their condition.

Optimised conservative management

The OCM package has been designed by pelvic floor and LARS experts based on current evidence and reviewed by PPI representatives.²³ Sites will be trained to deliver the OCM through Site Initiation Visits, a POLARiS-specific training video and support

	Events	Baseline ¹	1 month (post education session)	3 months	6 months	9 months	12 months	Continue 3 monthly ²	24 months
	Consent	✓							
Assessments/ Investigations	Randomisation	✓							
	Clinical Assessment	✓		✓					
mer gati	TAI Education session	✓							
stig	TAI Troubleshooting appt		✓						
lsse nve	Telephone Assessment				✓	✓	✓		
~ -	Clinical Data Collection (medical notes)								✓
	Adverse Events		✓	✓	✓	✓	✓		✓
	LARS Questionnaire	✓		✓	✓	✓	✓	√	✓
d d res	LARS ICAT	✓		✓	✓		✓		1
ipar lete inai	EQ-5D-5L	✓		✓	1		1		✓
m pl	EORTC CR29	✓		✓	1		✓		✓
Participant completed questionnaires	QLQ-C30	✓		✓	1		1		1
ō	Health Resource Use	✓		✓	*		*		1
ve4	Interview	√3		*					4
Qualitative⁴	Survey			~					~

Baseline participant completed questionnaires should be completed after consent but before randomisation

Figure 3 The timings of data collection points and clinical assessments for the randomised controlled trial patients randomised to TAI are summarised in the table below. LARS iCAT, low anterior resection syndrome Impact and Consequences Tool; TAI, transanal irrigation.

¹Baseline participant completed questionnaires should be completed after consent but before randomisation

² LARS score will continue to be completed every 3 months throughout the 24 months follow up period

³ Up to 6 weeks from randomisation

⁴A sub-set of participants only will take part in the qualitative sub-study interviews and surveys

² LARS score will continue to be completed every 3 months throughout the 24 months follow up period

⁴A sub-set of participants only will take part in the qualitative sub-study interviews and surveys

	Events	Baseline ¹	Operative (temporary SNM)	2-week post- operative review	Operative (permanent SNM)	3 months	6 months	9 months	12 months	Continue 3 monthly ²	24 months
	Consent	✓									
<u>~</u> ∞	Randomisation	✓									
tion	Clinical Assessment	✓	✓	✓	✓	✓					
ssm	Telephone Assessment						✓	✓	✓		
Assessments/ Investigations	Clinical Data Collection (medical notes)										✓
	Adverse events	✓	✓	✓	✓	✓	✓	✓	✓		✓
	LARS Questionnaire	✓		✓		✓	✓	✓	✓	✓	✓
d d	LARS ICAT	✓				✓	✓		✓		✓
ipar lete	EQ-5D-5L	✓				✓	✓		✓		✓
Participant completed uestionnaire	EORTC CR29	✓				✓	✓		✓		✓
Participant completed questionnaire	QLQ-C30	✓				✓	✓		✓		✓
	Health Resource Use	✓				✓	✓		✓		✓
ualitative ⁴	Interview	√ 3				√					*
Qualit	Survey					~					~

¹Baseline participant completed questionnaires should be completed after consent but before randomisation

Figure 4 The timings of data collection points and clinical assessments for the randomised controlled trial patients randomised to SNM are summarised in the table below. LARS iCAT, low anterior resection syndrome Impact and Consequences Tool; SNM, sacral neuromodulation.

document for healthcare professionals. The OCM package includes an initial appointment to establish which symptoms are most bothersome and the impact this has on the participants daily activities, followed by the provision of practical support and advice as well as tailored treatment options such as diet, medication and physiotherapy.

Practical support includes toilet access schemes, toilet positioning, bowel habit training and recommending support groups. Dietary advice includes completion of a food diary and onward referral to a dietician if available at that site. The clinician also completes a review of the current medication and will discuss medication options. All patients are encouraged to complete pelvic floor exercises and onward

referral to a pelvic floor physiotherapist is recommended if available at that site.

An additional POLARiS-specific OCM support document summarising the above information is provided to all participants randomised to OCM. Onward referral can be made to a dietician and pelvic floor physiotherapist if available at that site. Further advice and medication adjustments will be offered during the 3-month post-randomisation visit (figure 2).

The following interventions used in this study are Conformité Européenne (CE) and/or UK Conformity Assessed (UKCA) marked and being used within their licensing specification. Due to the nature of the interventions, this is a non-blinded trial.

		Timepoints post-registration							
	Events	Baseline	3 months	6 months	9 months	12 months	Continue 3 monthly ³	24 months	
	Clinical Assessment	√ 1				✓²		√2	
Assess ments/ Investig ations	Adverse Events					✓		✓	
	Cohort Consent	✓							
	Registration	✓							
_	LARS Score	✓	√	✓	✓	✓	✓	✓	
Participant completed uestionnair es	LARS ICAT	✓	√	✓		✓		✓	
	EORTC CR29	✓	✓	✓		✓		✓	
	QLQ-C30	✓	✓	✓		✓		✓	
		✓	✓	<u> </u>		· .			

¹Face to face or telephone assessment

Figure 5 The timings of data collection points and clinical assessments for the cohort are summarised in the table below. LARS iCAT, low anterior resection syndrome Impact and Consequences Tool.

² LARS score will continue to be completed every 3 months throughout the 24 months follow up period

³ Up to 6 weeks from randomisation

⁴A sub-set of participants only will take part in the qualitative sub-study interviews and surveys

 $^{^{\}mathbf{2}}$ Later clinical data collection will be collected from medical notes,

³ LARS score will continue to be completed every 3 months throughout the 24 months follow up period



TAI

TAI involves instilling warm water into the rectum and colon via the anus to empty out the stool. There are several commercially available TAI systems and broadly, systems can be divided into low and high volume. TAI can be delivered via a variety of different systems the choice of which is determined by a clinician. Each site will be responsible for the procurement of irrigation systems. Treatment involves a 1-hour education session with a nurse and provision of a starter pack. Participants will have a follow-up/troubleshooting review 1 month after the education session (figure 3).

SNM

Participants randomised to SNM will undergo a two-stage procedure. Initially a temporary device is fitted with a 2-week test phase. The response is assessed during the test phase and if deemed successful by the patient and clinician a permanent device is fitted (figure 4). Both procedures are performed as day-cases. Postoperative care will be as per each site's standard practice. The decision of which SNM device is used, and whether to use a temporary wire electrode or the tined quadripolar electrode lead for the test phase will be at the discretion of the clinical team/patient.

Follow-up

Cohort follow-up is over a 24-month period. Demographic information and relevant medical history will be collected at baseline for all cohort participants. Follow-up data will be collected from medical notes at 12 and 24 months post-registration. In addition, cohort patients will be asked to complete the LARS score on a 3-monthly basis throughout the 24-month follow-up period and a set of questionnaires (LARS Impact and Consequences Tool (iCAT), EORTC QLQ-C29 and EORTC QLQ-CR30) at baseline and at 3, 6, 12 and 24 months post-registration (figure 5).

RCT participants will complete the same set of questionnaires at the same time intervals as the cohort participants, with the addition of EQ-5D-5L and Health Resource Use Questionnaires. They will also have a series of clinical reviews at baseline and at 3, 6, 12 and 24 months postrandomisation in person, via telephone and clinical notes with some variation depending on which treatment the participant has been randomised to (figures 2–4). Participants for both cohort and RCT can choose to complete their questionnaires electronically or on paper. Should a participant be recruited to the RCT from the cohort, the cohort follow-up will cease, and the participant will be follow-up as per the RCT timelines only.

Sample size calculation

The minimal clinical important difference (MCID) is defined as a 5-point difference in LARS score. ²⁴ A conservative SD in LARS scores of 15 is assumed. ⁶ The primary outcome is LARS score over 24 months post-randomisation adjusting for baseline. The required sample size for the

adjusted analysis is obtained by multiplying the sample size requirement for the unadjusted analysis by a factor of $(1r^2)$, where r is the correlation between the 24 month and baseline scores, ²⁵ assumed to be a weak correlation of 0.3. Attrition is assumed to be no more than 15%.

The number of patients recruited through three randomisation options: (1) SNM versus TAI versus OCM (2) SNM versus OCM (3) TAI versus OCM are denoted n1, n2 and n3, respectively, hence (2/3)*n1+n2 patients and (2/3)*n1+n3 patients will contribute to the SNM versus OCM and TAI versus OCM comparisons, respectively. It is assumed that 50% of RCT participants will be entered via randomisation option (1), 25% via randomisation option (2), and 25% via randomisation option (3) hence for a total of N participants, 0.25*N + (2/3)*0.5*N will contribute to each of the two primary comparisons.

326 participants are sufficient to yield 85% power to detect the MCID in an (unadjusted) two-sided t-test of 24-month LARS scores at the 5% level of significance (nQuery v3.0). Applying baseline adjustment factor of $(1-0.3^2)=0.91$ and accounting for 15% attrition results in a sample size target of 350 participants for each comparison. Recruiting a total of 600 participants to RCT allows 350 patients needed for each primary comparison (0.25*600 + (2/3)*0.5*600). Including an additional 200 Australian participants inflates the power to over 90% power to detect the MCID.

OUTCOME MEASURES

Primary outcome measure (RCT and cohort study)

The primary outcome measure for both the cohort study and the RCT is the LARS score over 24 months. The LARS score is an internationally validated five-question assessment exploring different bowel dysfunction and their frequency. ²⁶ The overall score (maximum 42) corresponds to either no LARS (0–20), minor LARS (21–29) or major LARS (≥30). The LARS score will be collected at baseline and 3-monthly intervals until 24 months post-registration to cohort or randomisation.

Secondary outcome measures (RCT and cohort study)

Health-related QoL will be measured using the EORTC QLQ – C 29 and EORTC QLQ – CR 30, both internationally validated, cancer-specific QoL Questionnaires covering multiple domains including emotional and physical function. ^{27 28} The EORTC QLQ – CR29 specifically focuses on colorectal cancer.

The LARS iCAT is a new patient-reported outcome measure developed by an international group. The tool is based on the international consensus definition of LARS, involving a large international patient panel, which identified eight symptoms and eight consequences of LARS. LARS iCAT is intended as comprehensive tool to assess treatment response and enable patient phenotyping. The LARS iCAT will be collected as part of the questionnaire set with the aim of validating the tool before use in clinical practice.

Information on all adverse events with a causal relationship to the trial or trial interventions will be collected for this trial whether volunteered by the participant, or detected by investigator on questioning, physical examination or other investigations. This is for both cohort and RCT patients from the date of registration/randomisation throughout the 24-month follow-up period. Adverse events will be graded for severity.

Secondary outcome measures (RCT only)

Treatment compliance will be measured via appointment attendance, ongoing use of TAI and decision to have the permanent SNM device fitted. An additional QoL tool, the EQ-5D-5L, will be used to generate a single index value for health status which is valuable in the assessment of healthcare evaluation and economic analysis. ²⁹ A Health Resource Use Questionnaire is also included for the health economics assessment.

DATA ANALYSIS PLAN

All analyses will be pre-specified in a Statistical Analysis Plan according to published guidance. Analysis of the cohort data will include exploratory model-fitting to evaluate the longitudinal trends in LARS scores in relation to patient characteristics and clinical data, as well as the identification of potential risk factors for major LARS. Analyses of the RCT will be conducted on an intention-to-treat basis. Comparative effectiveness analyses for each endpoint will consist of two pairwise treatment comparisons: TAI versus OCM and SNM versus OCM. Treatment groups will be combined across different randomisation options for gains in efficiency. There is no interim analysis planned.

Primary analysis

For each of the pairwise comparisons (SNM versus OCM, and TAI versus OCM) the primary analysis plan is to estimate the expected difference in LARS scores at 24 months post-randomisation, using constrained longitudinal multi-level statistical models.³² Transformations of the LARS score and/or non-normal error assumptions will be considered as required based on the observed distribution of scores. The stratification factors 'time between anterior resection and randomisation', 'biological sex at birth', 'age', 'radiotherapy' and 'procedure' will be included as fixed effects in the model.

The model will account for nesting of repeated LARS score observations within patient, and the nesting of patients within 'Centre' within 'Country' by using appropriate variance components to model random effects. ³³ Randomisation option will also be accounted for in the model. ³⁴ Point estimates of treatment effects will be reported with two-sided 95% CIs.

Missing data mechanisms will be explored. Multiple Imputation using Chained Equations will be considered to explore the potential impact of missing data under a missing at random (MAR) assumption and to explore

sensitivity of the results to missing not at random (MNAR) mechanisms as required.

Secondary analysis

For all continuous secondary endpoints two pairwise comparisons will be made (TAI:OCM, SNM:OCM), with analysis populations and modelling approaches as already described for the primary outcome measure. Transformations of the outcomes and/or non-Normal error assumptions will be considered as required based on the observed distribution of outcomes. Questionnaire domains will be scored according to scoring manuals and reported graphically over time. Adverse events will be reported descriptively.

QUALITATIVE SUB-STUDY, PROCESS AND ECONOMIC EVALUATION

The qualitative sub-study, process and economic evaluation will be conducted as separate components in each country to reflect local differences. The Centre for Healthcare Evaluation, Device Assessment and Research (CEDAR) at C&V UHB is responsible for the design, implementation, management and analysis of the UK economic evaluation, process evaluation and qualitative sub-study. The NHMRC Clinical Trials Centre (CTC) at the University of Sydney will undertake an Australian health system cost-effectiveness evaluation. The Australian qualitative sub-study will be undertaken by the Surgical Outcome Research Centre (SOuRCe) at Royal Prince Alfred Hospital and the University of Sydney.

Qualitative sub-study

Few qualitative studies have explored patient perspectives on LARS treatment strategies.³⁵ The purpose of the qualitative sub-study was to explore the impact of the trial interventions on participants' (1) QoL, (2) activities of daily living, (3) LARS symptoms and (4) psychological functioning. Participants who are randomised into the RCT will be given the opportunity to take part in the qualitative sub-study.

In the UK a series of three semi-structured interviews will take place over the 24 months follow-up period. In Australia the three interviews will take place over 12 months. Demographic data will be used to purposively sample a broad range of participants. Up to 48 UK participants and 24 Australian participants will be interviewed. An interview topic guide will be formulated with input from PPI representatives and healthcare professionals (online supplemental files 2 and 3). Interviews will be carried out by a qualitative researcher who is independent of the participants care team.

Transcripts will be imported and coded using NVivo (QSR International). Analysis of the qualitative data will use a mainly iterative-inductive approach and further line-by-line coding will allow for identification of emerging themes. Results from this will then be integrated with the results from the main trial to fully and accurately reflect



patient experiences of LARS and the impact of the trial treatments. Data from Australian interviews will form a stand-alone publication.

Process evaluation

The purpose of the process evaluation was to provide context to the efficacy findings of the study. It will use mixed methods to investigate the domains of acceptability, implementation, including fidelity, dose and reach, mechanisms of impact and context, as defined by MRC guidance for process evaluations.³⁶ A mixed-methods approach will be taken using data from patient surveys (optional for patients, administered at 3-month and 24-month follow-ups), staff interviews (n=24 from 12 sites during the first and last 6months of recruitment), study records and clinical case report forms (CRFs). Staff interviews will be audio-recorded and used to create anonymised transcripts. These transcripts and qualitative data from survey responses will be analysed thematically using NVivo (QSR International).³⁷ Quantitative data will be analysed descriptively using a statistical package (SPSS 29). Qualitative and quantitative findings will be integrated at the conclusion of the study.

A process evaluation pilot report, using the pilot patient survey and staff interview data will be presented to the TMF following the internal pilot to facilitate the optimisation of study procedures for participants and staff.

Economic evaluation

The economic evaluation will include a within-trial economic analysis of patient-level trial data and a long-term decision analytic model to extrapolate long-term costs and outcomes of each treatment group. In both the UK and Australia, RCT participants will complete a Health Resource Use Questionnaire at baseline, 3, 6, 12 and 24 months post-randomisation. This will capture data such as primary, secondary and private healthcare resource use, personal expenses on consumables, medicines and travel costs. In Australia, RCT participants' use of medical services and prescription medicine in the primary care setting will be linked to Medicare and Pharmaceuticals Benefits data with Services Australia. CRFs will be used to capture patient-level resource use associated with the treatment during the trial period.

Effectiveness will be measured in terms of quality-adjusted life years. The primary analysis will evaluate cost-effectiveness from a healthcare payer perspective and a secondary analysis will be undertaken to include wider societal costs. Deterministic and probabilistic sensitivity analyses will be performed to characterise the parametric uncertainty associated with the cost and outcome differences between groups. The reporting of this economic evaluation in both the UK and Australia will comply with CHEERS 2022 recommendations.³⁸

DATA COLLECTION AND MANAGEMENT Data collection

Participating sites will be expected to maintain a file of essential trial documentation, which will be provided by the CTRU, and keep copies of all completed paper CRFs. Participant questionnaires will be collected either on paper or electronically. Trial data collected on paper CRFs will be submitted to the CTRU usually via standard post or secure electronic transfer. All other data collection will be via Remote Data Entry on electronic CRFs (eCRFs) onto a trial-specific database managed by the CTRU at the University of Leeds. Data will only be completed by personnel authorised to do so by the PI and recorded on the trial-specific Authorised Personnel log.

Confidentiality

Participant data collected during the trial will be kept strictly confidential and accessed only by delegated members of the research team. Copies of the UK participant consent forms will include participant names. Postal address, email address and phone number will also be collected depending on how a participant opts to complete their follow-up. All other data collection forms that are transferred to or from the CTRU will be coded with a unique participant trial number. Data will be held securely on paper and electronically at the CTRU for the duration of the trial. The CTRU will have access to the entire database for monitoring, co-ordinating and analysis purposes.

If a participant withdraws consent from further collection of data, the data collected up to the date of withdrawal will remain on file and will be included in the final trial analysis. Trial data will be retained for a minimum of 5 years. Data will be made available for secondary research once the main trial objectives are complete.

Trial management and monitoring

Trial supervision will be established according to the principles of Good Clinical Practice (GCP) and in line with the NHS UK Policy Framework for Health and Social Care. This includes a Trial Management Group (TMG), an independent Trial Steering Committee (TSC) and an independent Data Monitoring and Ethics Committee (DMEC).

The TMG, comprising the CI, CTRU team, CEDAR team, NHMRC CTC representatives, Australian Co-Chairs and other key external members of staff involved in the trial, and patient representatives have responsibility for the clinical set-up, ongoing management, promotion of the trial, and for the interpretation of results. The DMEC is appointed to review the safety and ethics of the trial, alongside trial progress and the overall direction overseen by the TSC. Trial progress will be closely monitored by the DMEC by detailed unblinded reports prepared by the CTRU. The independent TSC will provide overall supervision of the trial.

The trial includes an internal pilot phase, within the first 12 months of open recruitment. The internal pilot will assess recruitment, sites and qualitative evaluations. Following the 12-month internal pilot the DMEC and TSC will review trial progress to independently advise on progression of the trial to the funder.



Safety reporting

An 'adverse event' in POLARiS is defined as an untoward medical event in a participant that is related to the treatment of LARS. Examples of expected AEs in POLARiS include; wound-infection after SNM, pain or bleeding as a result of TAI and side effects of medications used for SNM. Adverse events will be graded for severity using CTCAE grading for all adverse events. Intra-operative adverse events will be graded using ClassIntra classification and postoperative adverse events will be graded using the Clavien-Dindo Classification scale. Any serious AEs or related unexpected serious AEs should be reported to the CTRU within 24 hours. It is the role of either the PI or CI to assign relatedness and expected nature of serious AEs. Both the DMEC and TSC perform periodic reviews of the safety data.

ETHICS AND DISSEMINATION

The trial will be performed in accordance with the principles of Good Clinical Practice and the Declaration of Helsinki (2013). This trial was reviewed and approved by Wales REC 4 (reference: 23/WA/0171) in the UK and Sydney Local Health District HREC (reference: 2023/ETH00749) in Australia. The appropriate local approval will be obtained by each participating site before entering participants into the trial. Further ethical approval of amendments to the protocol and/or related documents will be sought as necessary.

The trial outcomes will be disseminated to participants on request and published upon completion of the trial in a peer-reviewed journal and at international conferences. Credit for the main results will be given to all those who have collaborated in the trial. Authorship for the publication of the results of this study will be based on the principles of the International Committee of Medical Journal Editors Recommendation 2018. Data will be made available for secondary research once the main trial objectives are complete.

Author affiliations

¹Clinical Trials Research Unit, Leeds Institute of Clinical Trials Research, University of Leeds, Leeds, UK

²School of Medicine, Cardiff University, Cardiff, UK

³Colorectal Surgery, University Hospital of Wales, Cardiff and Vale University Health Board, Cardiff, UK

⁴Cardiff and Vale University Health Board, Cardiff, UK

⁵CEDAR, University Hospital of Wales Healthcare NHS Trust, Cardiff, UK

⁶Research and Development, Cardiff and Vale University Health Board, Cardiff, UK

⁷Cwm Taf Morgannwg University Health Board, Abercynon, UK

⁸NHMRC Clinical Trials Centre, University of Sydney CAR, Glebe, New South Wales, Australia

⁹Involving People Wales, Health and Care Research Wales, Cardiff, UK

¹⁰Department of Colorectal Surgery, Leeds Teaching Hospitals NHS Trust, Leeds, UK

¹¹Department of Colorectal Surgery, QEII Jubilee Hospital, Acacia Ridge, Queensland, Australia

¹²Department of Colorectal Surgery, Royal Brisbane and Woman's Hospital Health Service District, Herston, Queensland, Australia

¹³Department of Colorectal Surgery, Royal Prince Alfred Hospital, Camperdown, New South Wales, Australia ¹⁴Department of Colorectal Surgery, Concord Repatriation General Hospital, Concord, New South Wales, Australia

¹⁵Central Clinical School, The University of Sydney Faculty of Medicine and Health, Sydney, New South Wales, Australia

¹⁶Surgical Outcomes Research Centre, Royal Prince Alfred Hospital, Camperdown, New South Wales, Australia

¹⁷Centre for Academic Surgery, Barts and The London School of Medicine and Dentistry Blizard Institute, London, UK

¹⁸Consultant Colorectal Surgeon, Leeds Teaching Hospitals NHS Trust Centre for Respiratory Medicine, Leeds, UK

¹⁹Early Years Consultant Network, The Association of Coloproctology of Great Britain and Ireland, London, UK

X Emily Farrow @EZFarrow, Alexandra Harriet Coxon-Meggy @coxon_harriet, Neil Corrigan @NPCorrigan and Julie Cornish @jules cornish

Contributors JCr, NC, DS, MD, JW, LK, RP, JH, DJ, JT, AW, K-SN, KW, CK, AQ and JCo conceptualised the trial, obtained funding and wrote the trial protocol. AC-M, KG, HM, HYC, AM, CL, BT and AD were also involved in writing the trial protocol. All authors have contributed to the writing and review of this report which is based on Protocol Version 3.0 dated 05/03/2024. JCo is the guarantor for the contributorship statement.

Funding This work is supported by National Institute for Health Research (NIHR) Health Technology Assessment (HTA) Programme (Grant Ref: NIHR134937) in the UK and NHMRC-NIHR Collaborative Research Grant in Australia (Ref: 2015501). The views expressed are those of the author(s) and not necessarily those of the NIHR or the Department of Health and Social Care.

Competing interests All authors were required to provide ICMJE COI disclosure form, the following disclosures were made: DDS is fully funded through the NIHR as a Research Professor and is an NIHR Efficacy and Mechanism Evaluation Funding Committee member. CHK has carried out speaker and consultancy work for Medtronic and consultancy work for Exero Medical. He is Chief Medical Officer and co-founder of Amber Therapeutics, and CMO and shareholder at Enterika LTD. AQ is a member of ACPGBI Advanced Cancer sub-committee and has received a ZR MedTech Honoraria for lecturing at the Chinese Colorectal Society. JCo has received an educational grant from BD & Medtronic and is Clinical Director of Everywoman Festival. All other authors made no declarations.

Patient and public involvement Patients and/or the public were involved in the design, or conduct, or reporting, or dissemination plans of this research. Refer to the Methods section for further details.

Patient consent for publication Not applicable.

Provenance and peer review Not commissioned; externally peer reviewed.

Supplemental material This content has been supplied by the author(s). It has not been vetted by BMJ Publishing Group Limited (BMJ) and may not have been peer-reviewed. Any opinions or recommendations discussed are solely those of the author(s) and are not endorsed by BMJ. BMJ disclaims all liability and responsibility arising from any reliance placed on the content. Where the content includes any translated material, BMJ does not warrant the accuracy and reliability of the translations (including but not limited to local regulations, clinical guidelines, terminology, drug names and drug dosages), and is not responsible for any error and/or omissions arising from translation and adaptation or otherwise.

Open access This is an open access article distributed in accordance with the Creative Commons Attribution 4.0 Unported (CC BY 4.0) license, which permits others to copy, redistribute, remix, transform and build upon this work for any purpose, provided the original work is properly cited, a link to the licence is given, and indication of whether changes were made. See: https://creativecommons.org/licenses/by/4.0/.

ORCID iDs

Julie Croft http://orcid.org/0000-0001-7586-3394

Emily Farrow http://orcid.org/0000-0001-7498-1161

Alexandra Harriet Coxon-Meggy http://orcid.org/0000-0001-5637-8457

Katie Gordon http://orcid.org/0000-0003-1483-0657

Neil Corrigan http://orcid.org/0000-0002-1424-9830

Megan Dale http://orcid.org/0000-0001-7229-5109

Jared Torkington http://orcid.org/0000-0002-3218-0574

Charles H Knowles http://orcid.org/0000-0001-9854-6754

Julie Cornish http://orcid.org/0000-0003-4360-4472



REFERENCES

- 1 Bowel cancer statistics | cancer research UK. Available: https://www.cancerresearchuk.org/health-professional/cancer-statistics/statistics-by-cancer-type/bowel-cancer#heading-Zero [Accessed 29 Jul 2024].
- National bowel cancer audit an audit of the care received by people with bowel cancer in England and Wales the national bowel cancer audit (NBOCA). 2023.
- 3 Survival | bowel cancer | cancer research UK. Available: https://www.cancerresearchuk.org/about-cancer/bowel-cancer/survival [Accessed 30 Jul 2024].
- 4 Bryant CLC, Lunniss PJ, Knowles CH, et al. Anterior resection syndrome. Lancet Oncol 2012;13:e403–8.
- 5 Pape E, Pattyn P, Van Hecke A, et al. Impact of low anterior resection syndrome (LARS) on the quality of life and treatment options of LARS - A cross sectional study. Eur J Oncol Nurs 2021;50:101878.
- 6 Battersby NJ, Bouliotis G, Emmertsen KJ, et al. Development and external validation of a nomogram and online tool to predict bowel dysfunction following restorative rectal cancer resection: the POLARS score. Gut 2018;67:688–96.
- 7 Martellucci J, Sturiale A, Bergamini C, et al. Role of transanal irrigation in the treatment of anterior resection syndrome. *Tech Coloproctol* 2018;22:519–27.
- 8 Sun V, Wendel CS, Demark-Wahnefried W, et al. Diet and Behavior Modifications by Long-term Rectal Cancer Survivors to Manage Bowel Dysfunction-Associated Symptoms. *Nutr Cancer* 2019;71:89–99.
- 9 Food and Nutrition Board; Health and Medicine Division; National Academies of Sciences, Engineering, and Medicine. National academies of sciences, 'Examining access to nutrition care in outpatient cancer centers. In: Examining access to nutrition care in outpatient cancer centers. Washington, D.C: National Academies Press, 2016. Available: http://www.nap.edu/catalog/23579
- 10 Visser WS, Te Riele WW, Boerma D, et al. Pelvic floor rehabilitation to improve functional outcome after a low anterior resection: a systematic review. Ann Cologractal 2014;30:109–14.
- 11 Pelvic Floor Report | Practical recommendations to address inequalities in care for patients with pelvic floor disorders and to improve services for the future. 2024. Available: https://www. pelvicfloorreport.com/
- 12 Keane C, Fearnhead NS, Bordeianou LG, et al. International consensus definition of low anterior resection syndrome. ANZ J Surg 2020;90:300–7.
- 13 Koneru S, Cornish J, Chapuis PH, et al. Low anterior resection syndrome phenotypes-Different symptom profiles for different patients. Colorectal Dis 2024;26:1214–22.
- 14 Pieniowski EHA, Bergström CM, Nordenvall CAM, et al. A Randomized Controlled Clinical Trial of Transanal Irrigation Versus Conservative Treatment in Patients With Low Anterior Resection Syndrome After Rectal Cancer Surgery. Ann Surg 2023;277:30–7.
- 15 Rosen HR, Kneist W, Fürst A, et al. Randomized clinical trial of prophylactic transanal irrigation versus supportive therapy to prevent symptoms of low anterior resection syndrome after rectal resection. BJS Open 2019;3:461–5.
- 16 Jimenez-Gomez LM, Espin-Basany E, Marti-Gallostra M, et al. Low anterior resection syndrome: a survey of the members of the American Society of Colon and Rectal Surgeons (ASCRS. Int J Colorectal Dis 2016;31:813–23.
- 17 Huang Y, Koh CE. Sacral nerve stimulation for bowel dysfunction following low anterior resection: a systematic review and metaanalysis. *Colorectal Dis* 2019;21:1240–8.

- Marinello F, Fraccalvieri D, Planellas P, et al. Sacral Neuromodulation in Patients With Low Anterior Resection Syndrome: The SANLARS Randomized Clinical Trial. Dis Colon Rectum 2024;67:435–47.
- 19 National Guideline Alliance (UK). Optimal management of low anterior resection syndrome: colorectal cancer (update). London: National Institute for Health and Care Excellence (NICE). 2020.
- 20 Bibby AC, Torgerson DJ, Leach S, et al. Commentary: considerations for using the 'Trials within Cohorts' design in a clinical trial of an investigational medicinal product. *Trials* 2018;19.
- 21 Brown S, Thorpe H, Hawkins K, et al. Minimization--reducing predictability for multi-centre trials whilst retaining balance within centre. Stat Med 2005;24:3715–27.
- 22 Altman DG, Bland JM. Treatment allocation by minimisation. BMJ 2005;330:843.
- 23 Christensen P, IM Baeten C, Espín-Basany E, et al. Management guidelines for low anterior resection syndrome – the MANUEL project. Colorectal Dis 2021;23:461–75.
- 24 Colorectal cancer (update). 2020.
- 25 Clifton L, B.J., Clifton DA. Comparing different ways of calculating sample size for two independent means: A worked example. Contemp Clin Trials Commun 2019;13:100309.
- 26 Emmertsen KJ, Laurberg S. Low anterior resection syndrome score: development and validation of a symptom-based scoring system for bowel dysfunction after low anterior resection for rectal cancer. *Ann* Surg 2012;255:922–8.
- 27 Aaronson NK, Ahmedzai S, Bergman B, et al. The European Organization for Research and Treatment of Cancer QLQ-C30: a quality-of-life instrument for use in international clinical trials in oncology. J Natl Cancer Inst 1993;85:365–76.
- 28 Gujral Š, Conroy T, Fleissner C, et al. Assessing quality of life in patients with colorectal cancer: an update of the EORTC quality of life questionnaire. *Eur J Cancer* 2007;43:1564–73.
- 29 Devlin NJ, Brooks R. EQ-5D and the EuroQol Group: Past, Present and Future. Appl Health Econ Health Policy 2017;15:127–37.
- 30 Gamble C, Krishan A, Stocken D, et al. Guidelines for the Content of Statistical Analysis Plans in Clinical Trials. JAMA 2017;318:2337–43.
- 31 Howard DR, Brown JM, Todd S, et al. Recommendations on multiple testing adjustment in multi-arm trials with a shared control group. Stat Methods Med Res 2018;27:1513–30.
- 32 Liu GF, Lu K, Mogg R, et al. Should baseline be a covariate or dependent variable in analyses of change from baseline in clinical trials. Stat Med 2009;28:2509–30.
- 33 Browne WJ, Goldstein H, Rasbash J. Multiple membership multiple classification (MMMC) models. Stat Modelling 2001;1:103–24.
- 34 Lee KM, Turner RM, Thwaites GE, et al. The Personalised Randomized Controlled Trial: Evaluation of a new trial design. Stat Med 2023:42:1156–70.
- 35 McCutchan GM, Hughes D, Davies Z, et al. Acceptability and benefit of rectal irrigation in patients with Low Anterior Resection Syndrome: a qualitative study. Colorectal Dis 2017;20:76–084.
- 36 Moore GF, Audrey S, Barker M, et al. Process evaluation of complex interventions: Medical Research Council guidance. BMJ 2015;350:h1258.
- 37 Fereday J, Muir-Cochrane E. Demonstrating Rigor Using Thematic Analysis: A Hybrid Approach of Inductive and Deductive Coding and Theme Development. Int J Qual Methods 2006;5:80–92.
- 38 Husereau D, Drummond M, Augustovski F, et al. Consolidated Health Economic Evaluation Reporting Standards (CHEERS) 2022 Explanation and Elaboration: A Report of the ISPOR CHEERS II Good Practices Task Force. V Health 2022;25:10–31.